Author’s response to reviews

Title: Implementation of a Regional Quality Improvement Collaborative to Improve Care of People Living with Opioid Use Disorder in a Canadian Setting

Authors:
Laura Beamish (laura.beamish@gmail.com)
Zach Sagorin (zachsagorin@gmail.com)
Cole Stanley (cole.stanley@gmail.com)
Krista English (kristaenglish@icloud.com)
Rana Garelnabi (rgarelnabi@cfenet.ubc.ca)
Danielle Cousineau (Danielle.Cousineau@vch.ca)
Rolando Barrios (rbarrios@cfenet.ubc.ca)
Jan Klimas (jan.klimas@bccsu.ubc.ca)

Version: 3 Date: 09 Aug 2019

Author’s response to reviews:

Editor Comments

1. Consent to participate.
   Additional information has been added to the body of the manuscript to clarify the consent. Methods Section, Page 7, line 164. Participating teams consented by signing a letter of understanding before joining the Collaborative. The following information was also added to the Ethics approval and consent to participate section, Page 16, line 390; Collaborative teams consented by signing a letter of understanding before joining the Collaborative.

2. Availability of data and materials.
   Additional information has been added to the body of the manuscript to clarify the data collection. Methods Section, Page 8, line 195; Teams reported monthly aggregated, non-nominal quality metrics and qualitative description of change from October 2017 to December 2018. The following information was also added to the Availability of data and materials section, Page 17, line 393; The data that support the findings of this study are available from the corresponding author, RB, upon reasonable request. However, note that participating teams only reported aggregated data.
3. Clean manuscript.
The clean manuscript has been updated.